Author's response to reviews

Title: Ipsilateral Lower Extremity Joint Involvement Increases the Risk of Poor Pain and Function Outcomes after Hip or Knee Arthroplasty

Authors:

Jasvinder A Singh (Jasvinder.md@gmail.com)
David G Lewallen (Lewallen.david@mayo.edu)

Version: 2 Date: 15 January 2013

Author's response to reviews:

Dear Editor:
We have uploaded the strobe checklist and have updated the paper as well.
-jas singh